Author's response to reviews

Title: Validation of self reported diagnosis of hypertension in a cohort of university graduates in Spain

Authors:

Alvaro Alonso (aalonso@alumni.unav.es)  
Juan Jose Beunza (jjbeunza@unav.es)  
Miguel Delgado-Rodriguez (mdelgado@ujaen.es)  
Miguel Angel Martinez-Gonzalez (mamartinez@unav.es)

Version: 5 Date: 25 August 2005

Dear Editor,

Please find attached our revised version of the paper 'Validation of self reported diagnosis of hypertension in a cohort of university graduates in Spain' (MS: 9424857587124411), accepted for publication in BMC Public Health.

As requested, we have performed the following formatting changes:
- The title and abstract are identical in the manuscript and in the submission system.
- We are used sentence case in the title and in the section headings.
- We have removed authors' qualifications.
- We have removed keywords in the manuscript.
- We have changed abstract headings, as requested.
- We have removed highlighting and tracking.
- We have renamed the 'Introduction' as 'Background'.
- Figures legend and tables now are located after references.
- We now provide a higher resolution figure.

Do not hesitate to contact us if there are more changes requested.

Best,

Alvaro Alonso, MD, PhD
Department of Epidemiology. Harvard School of Public Health